Author’s response to reviews

Title: Apparent Diffusion Coefficient cannot predict molecular subtype and lymph node metastases in invasive Breast Cancer. A multicenter analysis

Authors:

Alexey Surov (alexey.surov@medizin.uni-leipzig.de)

Yun-Woo Chang (ywchang@schmc.ac.kr)

Lihua Li (lilh@hdu.edu.cn)

Laura Martincich (laura.martincich@ircc.it)

Savannah Partridge (scp3@u.washington.edu)

Jin Kim (youdosa@naver.com)

Andreas Wienke (andreas.wienke@uk-halle.de)

Version: 3 Date: 24 Oct 2019

Author’s response to reviews:

1. Please could you ensure only the authors of the paper are referred to in the author contributions section. Currently the initials HJM appear in the contributions but not in the author list.

We corrected the data as suggested.

2. Please provide clarification on whether any (administrative) permissions were required, and obtained, in order to review patient records and use the data mentioned in the Methods.

There were no permissions, which were required and obtained.

3. Multi-panel figures (those with parts a, b, c, d etc.) should be submitted as a single composite file that contains all parts of the figure.

It was corrected as suggested.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures
(and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Please find also the added clean copy of the manuscript.